Author's response to reviews

Title: Pharmacotherapy for Bipolar Disorder and concordance with treatment guideline: a tertiary care survey.

Authors:

   Sabrina Paterniti (sabrina.paterniti@rohcg.on.ca)
   Jean-Claude Bisserbe (Jean-Claude.Bisserbe@theroyal.ca)

Version: 2 Date: 2 December 2012

Author's response to reviews:

Dear Editor-In-Chief,

Please find enclosed a revised manuscript, which includes some modifications in the “Method” part, as requested. We have added the name of the ethics committee who approved the research carried out to collect the data analyzed in the paper. We have also mentioned that a written informed consent was obtained by the study participants, where the participant agrees that the results of their evaluation may be published, keeping individual identity completely confidential. A more detailed description of the selection method of the study was also added.

With my best regards,

Sincerely yours,

Sabrina Paterniti